Author's response to reviews

Title: Ovarian stromal tumor with luteinized cells: a rare recurrence of a rare tumor.

Authors:

TAZI EL MEHDI (moulay.elmehdi@yahoo.fr)
ISMAIL ESSADI (ismail@yahoo.fr)
HIND M’RABTI (hind@yahoo.fr)
HAASSAN ERRIHANI (hassan@yahoo.fr)

Version: 2 Date: 17 January 2011

Author's response to reviews:

1/ We modify the title page
2/ We restructured the abstract based on the journal recommendations
3/ We include the ethnicity of the patient
4/ We give a conclusion to the article
5/ We include the competing interest, authors contribution and patient consent

Sincerely

Dr TAZI EL MEHDI